Tue May 21 16:43:20 GVI 2002
Year: 02 Doy: 141
Qobserver: yasukawa
WEATHER COMMVENT: Tue May 21 16:44:18 GMIr 2002
Cool, clear, no w nd.

Tue May 21 16:49:49 GJVI 2002 CH P Start Patro
Tue May 21 16:50: 01 GVI 2002 MKI VvV Start Patro
Tue May 21 16:50: 06 GMI 2002 Pl CS Start Patro

**MLSO PROBLEMF*: Tue May 21 16:54:33 GMI 2002

UPS al arm annunci ator chirping when |I arrived. Message indicates battery
charger has failed. Systemcheck indicates batteries OK at this tine.

A NEW TAPE HAS BEEN PUT | NTO KAl EE DLT DRI VE, Tue May 21 16:55:48 GMI 2002
WEATHER COMMVENT: Tue May 21 17:01:57 GMIr 2002

Thi ck haze in saddle up to around 9000 ft, second |ayer at around 10, 000-11, 000
feet on Mauna Kea. Sky is very bright over Mauna Loa.

**MLSO PROBLEMF*: Tue May 21 17:39:10 GMr 2002

Hoi ke is only animating nk4 i nages and today’'s inmages are black & white with
no header info on them CH P just started animating, only the 19th’'s data
and one of today' s data so far, today' s data is black & white and al so

has no header on it. PICS inages are not animating

WEATHER COMMVENT: Tue May 21 17:42:03 GMr 2002

Rats! In cirrus from sout hwest.

**M.SO PROBLEMF*: Tue May 21 17:44:28 GV 2002

Hoi ke started animating PICS disk image with one of today’s inmages,

b&w, no header.

Tue May 21 18:02:25 GV 2002 Pl CS Fl at

Tue May 21 18:05:17 GVI 2002 Pl CS End Fl at

**M.SO PROBLEMF*: Tue May 21 18:30:35 GV 2002

PICS |inb ani mati ons now goi ng, no header, b&w. Alice is working on problem.
WEATHER COMMENT: Tue May 21 18:31:45 GMI 2002

Cirrus is getting thicker, stopping mnk4.

Tue May 21 18:37:16 GVI 2002 MKI V End Patro

Tue May 21 19:02:59 GVI 2002 CH P LSD

Tue May 21 19:05:15 GV 2002 CH P End LSD

Tue May 21 19:05:26 GVI 2002 CH P Bi asLSD

Tue May 21 19:06:21 GV 2002 CH P End Bi asLSD

Tue May 21 19:06: 30 GVI 2002 CH P Bi as

Tue May 21 19:07:23 GVI 2002 CH P End Bi as

WEATHER COMMVENT: Tue May 21 19:17:18 GMI 2002

There are a few hol es approaching, but the sky may still be very bright.

Restarting MK4 in case we hit some of those hol es.

Tue May 21 19:18:20 GVI 2002 MKI VvV Start Patro

GONG COMMVENT: Tue May 21 19:18:46 GMVI 2002

Going down to GONG to do weekly PMand to install a DLT drive.
WEATHER COMMVENT: Tue May 21 20:22:29 GMr 2002



I n orographic clouds.
GONG COMMENT: Tue May 21 20:22:52 GV 2002
Back from GONG
COVWMENT: Tue May 21 20:23: 07 GVI' 2002
Reconfi guring dome shutter.
Tue May 21 20:24:21 GV 2002 CH P End Patro
Tue May 21 20:24:41 GV 2002 Pl CS End Patro
Tue May 21 20:35:09 GVI 2002 MKI 'V End Patro
GONG COMMENT: Tue May 21 21:24:08 GMIr 2002
Goi ng down to GONG to check on CopyTape progress.
GONG COMMVENT: Tue May 21 21:34:49 GVII 2002
CopyTape appeared to be finished. Renoved data tapes and stored them on
the worktable for now as instructed. Installed tapes renoved to do
CopyTape back into the cart drives. Bad drive renoved by Darryl is
packed up but GONG asked that we wait before shipping it back. They will
advi se about the data tapes, the weekly maintenance | og that goes with the
data tape, and the bad drive.
WEATHER COMMVENT: Tue May 21 21:38:43 GMI 2002
O ographi ¢ clouds solid now.
Tue May 21 21:41:04 GV 2002
Ml V

16 _50.rawk4 17 _32.rawrk4 18 10.rawrk4 19 30.rawnk4 20_08.rawrk4
16 53.rawrk4 17 35.rawnk4 18 13.rawnk4 19 33.rawrkd4d 20 11.rawnk4
16 _56.rawrk4 17 38.rawnk4 18 16.rawrk4 19 36.rawrkd 20 _14.rawnk4
16 58.rawrk4 17 4l1l.rawnk4 18 19.rawrk4 19 39.rawrkd4d 20 _17.rawnk4
17_01l.rawrk4 17_43.rawnk4 18_22.rawrk4 19 _41.rawnk4 20_20.rawrk4
17 04.rawnkd4d 17 46.rawk4d 18 25.rawnkd 19 44.rawnkd4d 20 _23. rawk4
17 07.rawk4 17 _49.rawnk4 18 28.rawnrkd4d 19 47.rawnk4d 20 _26. rawk4
17 _10.rawrk4 17 52.rawk4 18 31.rawnrk4 19 50.rawrkd4d 20 _29.rawnk4
17 _16.rawrk4 17 55.rawnk4 18 34.rawnrk4 19 53.rawrkd4 20 31.rawnk4
17 _20.rawrk4 17 58.rawnk4 19 18.rawnk4 19 56.rawrk4
17_23.rawrk4 18 _01.rawrk4 19 21.rawrk4 19_59.rawrk4
17 26.rawk4 18 04.rawnk4 19 24.rawrkd4d 20 _02.rawnk4
17 29.rawk4 18 07.rawnk4 19 27.rawrk4 20_05.rawnk4



